Author's response to reviews

Title: Factors influencing the use of magnesium sulphate in pre-eclampsia/eclampsia management in health facilities in Northern Nigeria: a mixed methods study.

Authors:

Olugbenga Oguntunde (oguntundedoc@gmail.com)
Zulfiya Charyeva (zcharyeva@futuresgroup.com)
Molly Cannon (mcannon@futuresgroup.com)
William Sambisa (william.sambisa@gmail.com)
Nosakhare Orobaton (norobaton@tshipnigeria.org)
Ibrahim A Kabo (ikabo@tshipnigeria.org)
Kamil Shoretire (kshoretire@tshipnigeria.org)
Saba'atu E Danladi (sdanladi@tshipnigeria.org)
Nurudeen Lawal (nlawal@tshipnigeria.org)
Habib Sadauki (hsadauki@tshipnigeria.org)

Version: 4 Date: 1 April 2015

Author's response to reviews: see over
Dear Editor,
The description of how the single service provider per health facility was sampled has been rewritten in the method’s section of the manuscript.

This is reflected in line 136 - 139 as follows:

“For each facility, a complete list of service providers who rendered obstetric services was generated and using a table of random numbers, one service provider per health facility was randomly sampled for an interview.”

Thank you.

Olugbenga